Author's response to reviews

Title: Hypercalcemia in metastatic GIST caused by systemic elevated calcitriol: A case report and review of the literature.

Authors:

Katrine Hygum (katrhygu@rm.dk)
Christian N Wulff (chriwulf@rm.dk)
Torben Harsløf (torbhars@rm.dk)
Anders K Boysen (andboy@rm.dk)
Philip B Rossen (philross@rm.dk)
Bente L Langdahl (bente.langdahl@aarhus.rm.dk)
Akmal A Safwat (akmal@aarhus.rm.dk)

Version: 4
Date: 13 October 2015

Author's response to reviews: see over
Dear Krisha Mae Natan, Journal Editorial Office,

Thank you for the e-mail sent to us on October the 9th 2015. We are very pleased that our manuscript has been accepted for publication in BMC Cancer.

Please find enclosed a revised version of the manuscript “Hypercalcemia in metastatic GIST caused by systemic elevated calcitriol: A case report and review of the literature”. Changes made to the “Authors’ contributions” are marked in red, lines 190-192.

Yours sincerely

Katrine Hygum, MD
Department of Endocrinology and Internal Medicine
Aarhus University Hospital
8000 Aarhus C
Denmark